%%
%% Manuscript for Quark Matter 2012 Proceedings
%% ============================================
%% Author:
%% Collaboration:
%% Last updated:
%%
%% This is just a basic template. For more detailed options
%% see http://www.elsevier.com/wps/find/authorsview.authors/latex
%%
%%

%% This is the class option for the final document
\documentclass[final,1p,times]{elsarticle}
%% Use the option review to obtain double line spacing
%\documentclass[preprint,review,12pt]{elsarticle}

%% Useful/needed packages
\usepackage{graphicx}
\usepackage{amssymb}
\usepackage{amsthm}
\usepackage{lineno}


%% Your personal definitions go here
\newcommand{\pt}{$p_{T}$}
\newcommand{\pT}{$p_{T}$}
\journal{Nuclear Physics A}

\begin{document}
\begin{thebibliography}{00}

%\cite{Voloshin:2008dg}
%\bibitem{Voloshin:2008dg}
%  S.~A.~Voloshin, A.~M.~Poskanzer and R.~Snellings,
  %``Collective phenomena in non-central nuclear collisions,''
%  arXiv:0809.2949 [nucl-ex].
  %%CITATION = ARXIV:0809.2949;%%


%\cite{Hama:2004rr}
\bibitem{Hama:2004rr}
  Y.~Hama, T.~Kodama and O.~J.~Socolowski,
  %``Topics on hydrodynamic model of nucleus nucleus collisions,''
  Braz.\ J.\ Phys.\  {\bf 35}, 24 (2005).
%  [arXiv:hep-ph/0407264].
  %%CITATION = BJPHE,35,24;%%

%\cite{Gardim:2011re}
\bibitem{Gardim:2011re} 
  F.~G.~Gardim, Y.~Hama and F.~Grassi,
  %``Fluctuating Initial Conditions and Anisotropic Flows,''
  Prog.\ Theor.\ Phys.\ Suppl.\  {\bf 193}, 319 (2012).
%  [arXiv:1110.5658 [nucl-th]].
  %%CITATION = ARXIV:1110.5658;%%


%\cite{Petersen:2010cw}
\bibitem{Petersen:2010cw}
  H.~Petersen, G.~-Y.~Qin, S.~A.~Bass and B.~Muller,
  %``Triangular flow in event-by-event ideal hydrodynamics in Au+Au collisions
  %at $\sqrt{s_{\rm NN}}=200A$ GeV,''
  Phys.\ Rev.\  C {\bf 82}, 041901 (2010).
%  [arXiv:1008.0625 [nucl-th]].
  %%CITATION = PHRVA,C82,041901;%%

%\cite{Qin:2010pf}
\bibitem{Qin:2010pf} 
  G.~-Y.~Qin, H.~Petersen, S.~A.~Bass and B.~Muller,
  %``Translation of collision geometry fluctuations into momentum anisotropies in relativistic heavy-ion collisions,''
  Phys.\ Rev.\ C {\bf 82}, 064903 (2010).
%  [arXiv:1009.1847 [nucl-th]].
  %%CITATION = ARXIV:1009.1847;%%


%\cite{Qiu:2011iv}
\bibitem{Qiu:2011iv}
  Z.~Qiu and U.~W.~Heinz,
  %``Event-by-event shape and flow fluctuations of relativistic heavy-ion collision fireballs,''
  Phys.\ Rev.\ C {\bf 84}, 024911 (2011).
%  [arXiv:1104.0650 [nucl-th]].
  %%CITATION = ARXIV:1104.0650;%%

%\cite{Teaney:2010vd}
\bibitem{Teaney:2010vd} 
  D.~Teaney and L.~Yan,
  %``Triangularity and Dipole Asymmetry in Heavy Ion Collisions,''
  Phys.\ Rev.\ C {\bf 83}, 064904 (2011).
%  [arXiv:1010.1876 [nucl-th]].
  %%CITATION = ARXIV:1010.1876;%%

%\cite{Gardim:2011xv}
\bibitem{Gardim:2011xv} 
  F.~G.~Gardim, F.~Grassi, M.~Luzum and J.~-Y.~Ollitrault,
  %``Mapping the hydrodynamic response to the initial geometry in heavy-ion collisions,''
  Phys.\ Rev.\ C {\bf 85}, 024908 (2012).
%  [arXiv:1111.6538 [nucl-th]].
  %%CITATION = ARXIV:1111.6538;%%

%\cite{Alver:2010gr}
\bibitem{Alver:2010gr} 
  B.~Alver and G.~Roland,
  %``Collision geometry fluctuations and triangular flow in heavy-ion collisions,''
  Phys.\ Rev.\ C {\bf 81}, 054905 (2010)
  [Erratum-ibid.\ C {\bf 82}, 039903 (2010)].
%  [arXiv:1003.0194 [nucl-th]].
  %%CITATION = ARXIV:1003.0194;%%

\end{thebibliography}


\begin{frontmatter}

% Your Title - please insert
\title{Characterizing the hydrodynamic response to the initial conditions}% in event-by-event hydrodynamics}


%% Single author (and collaboration) - please insert
%\author{Gaius Julius Caesar (for the EMPIRE\fnref{col1} Collaboration)}
%\fntext[col1] {A list of members of the EMPIRE Collaboration and acknowledgements can be found at the end of this issue.}
%\address{St. Angelos Castle, Rome}

%% Multiple authors
\author[auth1]{Fernando G.  Gardim}
\author[auth1]{Fr\'ed\'erique Grassi}
\author[auth2,mcgill,lbnl]{Matthew Luzum}
\author[auth2]{Jean-Yves Ollitrault}
\address[auth1]{Instituto de F\'\i sica, Universidade de S\~ao Paulo, C.P. 66318, 05315-970, S\~ao Paulo-SP, Brazil}
\address[auth2]{CNRS, URA2306, IPhT, Institut de physique th\'eorique de Saclay, F-91191
Gif-sur-Yvette, France}
\address[mcgill]{McGill University,  3600 University Street, Montreal QC H3A 2TS, Canada}
\address[lbnl]{Lawrence Berkeley National Laboratory, Berkeley, CA 94720, USA}
%
\begin{abstract}

In hydrodynamics, the momentum distribution of particles at the end of the evolution is completely determined by initial conditions. We study quantitatively to what extent anisotropic flow $v_n$ is determined by predictors such as the initial eccentricity $\varepsilon_n$ in a set of realistic simulations, and we also show the importance of nonlinear terms in order to correctly predict $v_4$. This knowledge will be important for making a more direct link between experimental observables and hydrodynamic initial conditions, the latter being poorly constrained at present.  

\end{abstract}

\end{frontmatter} % do not change

%% linenumbers are useful for reviewing process
%\linenumbers

\section{Introduction}

One of the most important probes of ultrarelativistic heavy-ion collisions is the anisotropic flow. Event-by-event hydrodynamics \cite{Hama:2004rr} provides a natural way of studying anisotropic flow and its fluctuations.  In such calculations, one supplies a set of initial conditions (IC), evolves them through ideal or viscous hydrodynamics, and finally computes particle emission at the end. One can then write the azimuthal distribution of outgoing particles in a hydrodynamic event as a Fourier series
%
\begin{eqnarray}
\frac{2\pi}{N}\frac{dN}{d\phi}=1+2\sum_n^\infty v_n\cos[n(\phi-\Psi_n)],
\label{eq:v-def}
\end{eqnarray}
%
or equivalently $\left\{e^{i n\phi}\right\}=v_ne^{-i n\Psi_n}$, where $\{\cdots\}$ is the average in one event. Since the largest source of uncertainty in these calculations is the initial conditions,
it is useful to identify which properties of the initial state determine each observable.  This knowledge can then allow to constrain the initial state directly from data.
%, nevertheless anisotropic flow and event-plane angles are determined by these IC. The usual procedure is to choose one or two of models for the initial conditions and calculate the resulting flow observables, and compare the results with experimental data. 
%In order to constrain the initial state directly from data, we need to identify which properties of the initial state determine a given observable.

It is well known that the initial average profile of non-central collisions is almond-shaped. One can obtain the eccentricity and the direction of the initial profile computing $\varepsilon_2e^{i2\Phi_2}=-\left\{r^2e^{i2\phi}\right\}/\left\{r^2\right\}$ in the CM frame, where $\varepsilon_2$ is the participant eccentricity, $\Phi_2$ is the participant plane, whose direction is  the minor axis of the ellipse, and $\{\cdots\}$ denotes an average over the initial density profile. The hydrodynamic expansion produces an elliptic distribution of particles, i.e. $\varepsilon_2\propto v_2$, and the  maximum of the distribution of particles is aligned with the direction of the steepest energy gradient in the IC, i.e $\Phi_2=\Psi_2$. Nevertheless, due to the finite number of nucleons colliding in each nucleus-nucleus collisions, there are fluctuations in the initial profile, thus the relations $v_2\propto\varepsilon_2$ and $\Psi_2=\Phi_2$ may no longer be valid in event-by-event hydrodynamics. However, the study of these relations for several models of IC in event-by-event hydrodynamics, showed that they are reasonably satisfied \cite{Gardim:2011re,Petersen:2010cw,Qin:2010pf,Qiu:2011iv}.

%However, the scatter plot of $v_2$ as a function of $\varepsilon_2$, and the distribution of $\Psi_2-\Phi_2$ for different models of initial conditions in %event-by-event hydrodynamics, show that these relations are reasonable satisfied \cite{Gardim:2011re,Petersen:2010cw,Qiu:2011iv}.

%As for higher harmonics $v_n$, 
Symmetry considerations have been used to argue that higher-order flows ($v_n$ with $n\ge 3$) should also be created by an anisotropy, $\varepsilon_n$. It has been shown that this is also true for $n=3$, but it is not valid for $n=4,5$~\cite{Qin:2010pf,Qiu:2011iv}. Nevertheless, Teaney and Yan \cite{Teaney:2010vd} have introduced a cumulant expansion in the initial density profile, in which $\varepsilon_n$ is only the first term in an infinite series, and they have suggested that the hydrodynamic response may be improved by adding higher-order or nonlinear terms. 

In order to understand which properties of the initial state determine $v_n$ and $\Psi_n$, 
%and then ruled out a particular model of initial conditions if results do not match experimental data, 
we propose a simple quantitative measure of the correlation between $\left(v_n,\Psi_n\right)$ and any proposed predictor, such as $\left(\varepsilon_n,\Phi_n\right)$. We then use this to find better estimators for the anisotropic flows.  Details can be found in \cite{Gardim:2011xv}.
%
\section{Characterizing the hydrodynamic response}

In previous works~\cite{Petersen:2010cw,Qin:2010pf,Qiu:2011iv}, the correlation of the anisotropic flow with the initial geometry has been studied by visually inspecting two types of plots representing separately magnitude and direction: the $\Psi_n-\Phi_n$ distribution, and a scatter plot of $v_n$ versus $\varepsilon_n$. Instead, we carry out a global analysis, studying the correlation between the entire vector both simultaneously and quantitatively. For a given event in a centrality class, we write

\begin{equation}
v_n e^{in\Psi_n}=k\varepsilon_ne^{in\Phi_n}+\cal E,
\label{eq:estimator}
\end{equation}
%
where $k$ is a proportionality constant, and $\cal E$ is the difference between the calculated flow and the proposed estimator, or the error in the estimate.
% - first term in rhs of Eq. (\ref{eq:estimator}). 
 The generalized eccentricity $\varepsilon_n$, used throughout this work, is: $\varepsilon_ne^{i n\Phi_n}=-\left\{r^ne^{i n\phi}\right\}/\left\{r^n\right\}$,
%
%\begin{equation}
%\varepsilon_ne^{i n\Phi_n}=-\frac{\left\{r^ne^{i n\phi}\right\}}{\left\{r^n\right\}}
%\label{eq:edef}
%\end{equation}
%
which was proposed in Ref. \cite{Petersen:2010cw}, and later showed to be the lowest term in a cumulant expansion of the initial energy density \cite{Teaney:2010vd}. The best estimator is defined as the one that minimizes the mean square error $\langle|{\cal E}^2|\rangle$, where $\langle\cdots\rangle$ is the average over events in a centrality class. Thus, the best linear fit is achieved when 

\begin{equation}
k=\frac{\langle\varepsilon_n v_n\cos[n(\Psi_n-\Phi_n)]\rangle}{\langle\varepsilon_n^2\rangle}.
\label{eq:k}
\end{equation}
%
%Inserting Eq. (\ref{eq:k}) in (\ref{eq:estimator}) and taking the average over events in a centrality class, one obtains the mean-square error $\langle|{\cal E}^2|\rangle=\langle v_n^2\rangle-k^2\langle\varepsilon_n^2\rangle$. From these equations, 
We define the quality of the estimator by 
%\begin{equation}
%\langle|{\cal E}^2|\rangle}=\langle v_n^2\rangle-k^2\langle\varepsilon_n^2\rangle,
%\label{eq:mean-e}
%\end{equation}
\begin{equation}
Quality=k\frac{\sqrt{\langle\varepsilon_n^2\rangle}}{\sqrt{\langle v_n^2\rangle}}.
\label{eq:quality}
\end{equation}
%
The closer $Quality$ to 1, the better the response and the smaller the rms error $\langle|{\cal E}^2|\rangle$. Note that a negative $Quality$ means that $\Psi_n$ and $\Phi_n$ are anticorrelated. 
%As you will see in the next section, the anticorrelation appears in non-central collisions for $n=4$ and $5$.

%\subsection{Results}

We now present results for Au-Au collisions at the top RHIC energy using the hydrodynamic code NeXSPheRIO~\cite{Hama:2004rr}. NeXSPheRIO evolves initial conditions generated by the event generator NeXus, solves the equations of relativistic ideal hydrodynamics, then emits particles at the end of hydrodynamical evolution using a Monte-Carlo generator. For this work, we  generated 150 events in each of the $10\%$ centrality classes, and we added 115 events in an extra class for events with zero impact parameter - particles from the pseudorapidity interval $|\eta|<1$ were used. Eccentricities are obtained by averaging over the initial transverse energy density profile at z=0. Our hydrodynamical calculations contain fluctuating initial flow, as well as longitudinal fluctuations, thus the final flow measured is not entirely determined by the initial transverse geometry. In this sense, these results represent something of a worst-case scenario.

Fig. \ref{fig:1} displays the \textit{Quality} of the hydrodynamic response for elliptic and triangular flow as a function of centrality. As expected, the elliptic flow is driven to the almond-shaped overlap area for non-central collisions, where $\Psi_2$ is approximately coincident with $\Phi_2$. Even for central collisions, where $v_2$ comes from fluctuations, the best estimator given by Eq. (\ref{eq:estimator}) is able to capture the physics of the elliptic flow fluctuations. The \textit{Quality} for $n=3$ is not as close to 1 as for the elliptic case, but one can see that the triangularity $\varepsilon_3$ is still a good estimator of the triangular flow.
%, where the latter is due to fluctuations. 
We have also tested the triangularity definition proposed by Roland and Alver \cite{Alver:2010gr} (not shown, see Ref. \cite{Gardim:2011xv}), where $r^3$ is replaced by $r^2$ in Eq. (\ref{eq:estimator}). The triangularity with $r^3$ weight is a better predictor than with $r^2$ for most centralities. Since the $r^3$ weight gives more weight to outer shells of the initial energy profile, this means that the source of anisotropic flow moves inward as the collision becomes more peripheral.

\begin{figure}[h]
\begin{center}
 \includegraphics[width=0.58\textwidth]{v23.eps}
\end{center}
\vspace{-.7cm}
\caption{\textit{Quality} of the hydrodynamic response for elliptic (circle), triangular (square), quadrangular (triangle) and pentagonal flow (star) for different centrality bins. The leftmost points correspond to events with zero impact parameter.}
\label{fig:1}
\end{figure}
%
The estimators for $n=4,5$ differ qualitatively from our results for $v_2$ and $v_3$, where  $\varepsilon_4$ and $\varepsilon_5$ only give reasonable predictions for central collisions. Moving to peripheral collisions the \textit{Quality} decreases and even become negative. 
%In fact, this happen because $\Psi_4$ is anticorrelated with $\Phi_4$, as well as for $n=5$ \cite{Qiu:2011iv}. 
%(negative=anticorrelated, already written above)
This shows that $\varepsilon_n$ alone cannot be used to map the hydrodynamics response to the initial geometry for $n=4,5$.
%
\section{Finding better estimators}

One can hope to improve the estimator adding more terms in Eq. (\ref{eq:estimator}), e.g.

\begin{equation}
v_n e^{in\Psi_n}=k\varepsilon_ne^{in\Phi_n}+k'\varepsilon_n'e^{in\Phi_n'}+\cal E,
\label{eq:better}
\end{equation}
%
where $\varepsilon_n'$ and $\Phi_n'$ are other quantities determined from the initial density profile (for instance, the next higher cumulant). The procedure to obtain the better estimator is similar: minimize the mean-square error  $\langle|{\cal E}^2|\rangle$ with respect to $k$ and $k'$, take the average over events in each centrality class, then insert the values of $k$ and $k'$ in Eq. (\ref{eq:better}) and compute% the \textit{Quality}

\begin{equation}
Quality=\frac{\langle|k\varepsilon_ne^{in\Phi_n}+k'\varepsilon_n'e^{in{\Phi_n'}}|^2\rangle}{\langle{v_n^2}\rangle}.
\label{eq:nq}
\end{equation}
%
One can show that the quality is higher than with just one term, Eq. (\ref{eq:quality}). 

We want to apply the improved estimator to the quadrangular flow (for other flows see Ref. \cite{Gardim:2011xv}), but we need to know which are $\varepsilon_n'$ and $\Phi_n'$. Taking an elliptic initial condition, i.e an elliptic profile with only $\varepsilon_2\neq 0$, then evolving it through ideal hydrodynamics, and computing the particle emission, one will obtain, using Eq. (\ref{eq:v-def}), the anisotropic flows and the event-plane angles. Since there is only $\varepsilon_2\neq0$ one should expect only $v_2\neq0$. The results for this initial condition are: $v_2\neq0$ as a response to $\varepsilon_2$ and $\Psi_2=\Phi_2$, $v_n=0$ when $n$ odd (by symmetry), but there is a quadrangular flow, $v_4\neq0$, and $\Psi_4$ is aligned with $\Phi_2$. Since there is no $\varepsilon_4$, so where does $v_4$ come from? The answer is, $v_4$ is induced by the almond shape.

As quadrangular flow can also come from $\varepsilon_2$, we want to investigate, first, $v_4$ solely as a response to the almond shape. Thus, the natural estimator is $v_4 e^{i4\Psi_4}=k(\varepsilon_2e^{i2\Phi_2})^2+\cal E$ -- the square in the first term of rhs is used to preserve the rotational symmetry. The result can be seen in the Fig. \ref{fig:2}. For mid-central collisions, where $\varepsilon_2$ is large, the nonlinear term is important, yet this estimator is not as good as previous estimators of $v_2$ and $v_3$. Therefore, to obtain a better estimator for $v_4$ is necessary to combine both contributions, thus using Eqs. (\ref{eq:better}) and (\ref{eq:nq}), Fig. \ref{fig:2} show that the resulting estimator is good for all centralities. For $v_5$, the best estimator has linear e nonlinear terms: $\varepsilon_5$ and $\varepsilon_2\varepsilon_3$ \cite{Gardim:2011xv}.\\
%
\begin{figure}[htbp]
\begin{center}
 \includegraphics[width=0.6\textwidth]{v2.eps}
\end{center}
\vspace{-.7cm}
\caption{\textit{Quality} for 3 different estimators of $v_4$. The combined estimator (square) is the best estimator.}
\label{fig:2}
\end{figure}
%
%\section{Summary}
We have defined a quantitative measure of the quality of estimators of $v_n$ from initial conditions in event-by-event hydrodynamics. $v_2$ and $v_3$ are well predicted by $\varepsilon_2$ and $\varepsilon_3$, but for $v_4$ and $v_5$ it is necessary nonlinear terms. These results provide an improved understanding of the hydrodynamic response to the initial state in realist heavy-ion collisions.

%\section{Acknowledgments}
\vspace{.2cm}
\textbf{Acknowledgments:} This work is funded by FAPESP under projects 09/50180-0 and 09/16860-3, by CNPq under project 301141/2010-0, by “Agence Nationale de la Recherche” under grant ANR-08-BLAN-0093-01, and by Cofecub under project Uc Ph 113/08;2007.1.875.43.9. 


\section*{References}


\end{document}
\bye
